Author's response to reviews

Title: Obesity prevalence from a European perspective: a systematic review

Authors:

Anne Berghoefer (anne.berghoefer@charite.de)
Tobias Pischon (pischon@dife.de)
Thomas Reinhold (thomas.reinhold@charite.de)
Caroline M Apovian (caroline.apovian@bmc.org)
Arya M Sharma (sharma@ccc.mcmaster.ca)
Stefan N Willich (stefan.willich@charite.de)

Version: 5 Date: 23 May 2008

Author's response to reviews:

Typing errors in the figure B have been made.